The impact of income smoothing ¹ on earnings quality in emerging markets

Evidence from GCC markets

299

Evidence from GCC markets

Moade Fawzi Shubita School of Management/Accounting and Finance, New York Institute of Technology, Amman Campus, Jordan

Abstract

Purpose – The purpose of this paper is to assess the practice of income smoothing in the Gulf Cooperation Council (GCC) emerging markets; Saudi Arabia, Kuwait, United Arab Emirates, Oman and Qatar. Then, to examine the impact of income smoothing on the earnings quality to decide whether income smoothing can serve as either a tool to enhance earnings quality or a tool for opportunistic behavior. Audit quality and corporate governance as additional factors are considered in this study. Design/methodology/approach - The study methodology measures income smoothing behavior based on the coefficient of variation method. Earnings quality is measured as an outcome of the explained variations in stock returns by earnings based on the efficient market hypothesis. Audit quality is measured based on brand as higher quality assigned to auditor from any of the Big 4, while the corporate governance is addressed based on the extent of governmental ownership. The initial study sample comprises 55 companies over a ten year period, from 1999 to 2008; the final sample represents approximately 64 percent of the industrial sector that have public data during the study. **Findings** – The results suggest that income smoothing behavior in the GCC markets has many variations in practice. Income smoothing, on average, improves earnings quality in three countries out of four, but not significantly for the whole sample based on earnings level. The earnings changes model demonstrated a positive and significant impact of income smoothing on earnings quality. Audit quality and earnings quality have a positive relationship within the region, and companies dominated by the government perform well in accordance with the earnings-return model.

Research limitations/implications - The study is limited to the industrial sector of the GCC.

Practical implications – The study opens the door to future applications to other sectors within the GCC, same sectors and other sectors for Middle East countries and other emerging markets.

Social implications – The study may foster a better understanding of accounting practices in the GCC and Middle East. The study reveals variations in different aspects among GCC countries, this matter should be considered in separate studies across different areas.

Originality/value – The study makes an original contribution to being the first to explore this topic in the GCC. Additionally, this study shows that the GCC markets have different characteristics in the practice and impact of income smoothing on earnings' quality. Further, audit quality and corporate governance was investigated for each country and for the region, in addition to the interaction between these factors with the income smoothing and earnings quality.

Keywords Audit quality, Corporate governance, Emerging markets, Earnings quality, Income smoothing, Market-based accounting research

Paper type Research paper

1. Introduction

Several studies in the accounting literature conducted in several countries and based on various data sets have reached mixed results concerning the relationship between accruals and earnings ability to measure the company performance.



JEL Classification — G140, G340, M410, M420, M480, N250

An accounting classification of aggregate accruals results in the standard categories of purchases and sales of fixed assets, new finances raised from and payments to capital providers, long-term operating accruals and changes in working capital. A behavioral classification reflects the recognition that certain accruals are mandatory, arising naturally as part of the accounting reporting process, while other accruals are decided, both in their timing and magnitude, by managers. This produces a division of accruals into non-discretionary and discretionary components. The discretionary aspect of accruals can be used in smoothing earnings.

In the literature, there are two conflicting points of view regarding the consequence of income smoothing. One group supports smoothing as a valuable property of accounting earnings[1]. Another point of view is that smoothing is a poor property of accounting earnings that enables managers to satisfy their own interests by withholding valuable information[2].

This study will help to present the practice of income smoothing, earning quality relationship with income smoothing, corporate governance and audit quality factors in the Gulf Cooperation Council (GCC) region. Further, this study will try to uncover more facts about GCC markets based on the market-based accounting approach and will present more understanding of the extent of earnings smoothing, nature of earnings-return relationship and potential impact of additional factors; namely, audit quality and corporate governance.

GCC region has potential advantages and it deserves more attention from empirical studies that help achieve better understanding of this important region. There are several reasons that make the GCC region interesting area for conducting research. It provides a tax-free environment; this fact expects to attract investors' attention and will encourage external investments to explore this region. Despite this advantage, there are various studies that demonstrate a lack of external investment in the GCC region. Hammoudeh and Choi (2006) show that the major sources of variations in GCC return index returns come from the same region and global factors, in general, account for a small percentage of GCC stock markets' total variation. Moreover, Al-Shammari *et al.* (2008) show that GCC countries are still in the early stage of attracting foreign investors, as the median for internationality factor was only 0.01.

GCC countries have a geographic advantage over other regions as a result of their proximity to the world's major markets. A measure of this proximity can be derived from the concept of economic distance Behar and Manners (2010). Applying this measure, MENA region (which includes the GCC, Maghreb and the Levant) clearly has a lower economic distance figure than other emerging markets, such as East Asia. In addition, GCC countries currently hold about significant portion of the world's oil reserves, giving them an advantage over other emerging markets in terms of resource availability at the disposal of their respective governments.

GCC has strong cultural ties and similarities in many aspects, such as informal behavioral norms, values, customs and traditions. These similarities make it relatively convenient to navigate the different market opportunities in these countries. Conversely, there are challenges facing GCC countries, such as enhancing non-oil sectors, harmonizing implementation procedures to reflect similarities in the plans, directing research to serve market needs and enhancing transparency in various fields, such as foreign investment. In Summary, GCC countries are well placed to capitalize on their advantages and become a central part of tomorrow's global economy.

The current study will add to the literature in different ways. It will base its analysis on the industrial sector in GCC countries, and results will be provided at



JAEE

GCC and country level. Previous studies in this area either focus on other topics, Evidence from other sectors or just general approaches to similar topics, while this study will analyze in-depth this topic and add more factors, such as audit quality and corporate governance. This will make the study an advanced attempt to address an important issue in the GCC markets that have not received substantial attention in the literature. It will help achieve better understanding of the practices of income smoothing, audit quality and corporate governance in GCC, and will allow interaction between different variables for better understanding of the results. The approach adopted in the next sections is the general presentation of topic followed by addressing the topic in relation to GCC to enable better understanding for the reader and open the door for further research in the region. Further, this study could be extended to cover other emerging markets in the MENA, or any other, region.

The remainder of this paper is organized into eight sections. Section 2 presents a brief description of the GCC emerging markets and economy, Section 3 is a literature review, Section 4 addresses the process of detecting income smoothing, Section 5 presents the hypothesis of the study. Section 6 presents the sample of the study and descriptive statistics, Section 7 concerns hypothesis testing and inferential statistics and Section 8 contains conclusions and recommendations.

2. Brief description of the GCC emerging markets and economy

As the study period ends in 2008, this section offer a macroeconomic overview of the GCC only to the end of 2008, taking information from the GCC economic overview presented on the GulfBase web site (www.gulfbase.com) and the sixth and seventh editions of the Gulf Investment Guide for 2009 and 2010 produced by Zughaibi and Kabbani Financial Consultants. The GCC is an oil-based region with the largest proven crude oil reserves in the world, accounting for about 35.7 percent of the world's total in 2010. It is important to be aware about of this fact about the emerging markets in the GCC, as they represent a significant force in the international economy, are located in a very sensitive location, and play a very important role in the international political system. The principal macro indicators for the GCC are presented in Figures 1 and 2.

3. Literature review

3.1 General studies

Income smoothing has been defined by Beidleman (1973) as "dampening of fluctuations about some level of earnings that is currently considered to be normal for a firm[3]." Eckel (1981), Albrecht and Richardson (1990) and others distinguish between intentional and natural income smoothing. Natural income smoothing involves technical automatisms of accrual, while intentional smoothing can occur either by timing real business decisions (this is called real smoothing)[4] or by choosing accounting methods that allocate earnings over time in the desired manner (this is called as artificial smoothing). Horwtiz (1977) argues that real smoothing affects cash flow while artificial smoothing does not have such an impact.

The following figure present different types of income smoothing (Eckel, 1981) (Figure 3).

The question of the impact and intention of income smoothing remains controversial in the literature. Gassen et al. (2006) present two theories of income smoothing. On one hand, income smoothing can be viewed as having a signaling role in efficient capital



301

GCC markets



markets. In this sense, income smoothing is a vehicle for the disclosure of private information about future earnings (Barnea *et al.*, 1976). However, if the assumption of the efficient market is dropped, then income smoothing can be viewed as a tool utilized by managers to mislead market participants regarding future earnings or risk levels



for a company in order to lower the cost of capital and extract private benefits from Evidence from the company (Gordon, 1966). Thus, based on this argument the motivation and GCC markets expectations of consequences remain unclear.

Other studies have asserted a more definite vision of the role of income smoothing. Tucker and Zarowin (2006), for instance, address the question "Does Income Smoothing Improve Earnings Informativeness?" In their study, they define income smoothing in terms of the "amount of information about future earnings and cash flows that is reflected in current period stock returns." Their results show that a change in the current stock price of higher-smoothing firms contains more information about their future earnings than does a change in the stock price of lower-smoothing firms. Thus, they imply that managers use income smoothing to reveal private information about the firm's future profitability. Results indicate the same trend under several sensitivity tests. In the same vein, Cahen et al. (2008) explore the relationship between earnings informativeness and income smoothing under investor protection in 44 countries from 1993 to 2002[5]. Results show that earnings informativeness is more positively correlated with income smoothing in countries with strong investor protection. On the other hand, results show that in weak protection countries, managers are most likely to adopt income smoothing for opportunistic reasons. The overall conclusion is that the role of income smoothing "accounting discretion" is influenced by a country's overall institutional infrastructure mainly in the ability to provide protection for shareholders[6].

Other studies in this field address certain factors that may motivate the income smoothing. Hall and Frickel (2008) investigate the relationship between executive compensation and income smoothing. They conclude that compensation provides incentive for managers to smooth the time series of the reported earnings. Tseng and Wen Lai (2007) explore the relationship between income smoothing and profitability based on data on 142 companies listed on the Taiwan Stock Exchange for ten years from 1995 to 2004. Results show a negative relationship between income smoothing and profitability. Managers of companies facing lower profits have more incentive to smooth earnings according to the main findings of this study.

In addition to the aforementioned studies there are studies in the literature that investigate the consequences of income smoothing. Li and Richie (2009) adopt the approach of Tucker and Zarowin (2006) in determining the effect of income smoothing. Results show that smoother companies have a lower cost of debt and higher credit ratings. Additionally, results indicate that income smoothing serves companies in reducing the cost of capital. Further, Ayoib and Nooriha (2009), based on Malaysian data from 1991 to 2000, conclude that the existence of non-executive directors and the presence of brand name auditors was significant in delaying management from indulging in income smoothing. Ayoib and Nooriha (2009) adopt the coefficient of variation approach to detect income smoothing.

In the accounting literature, famous studies have generated well-known models to detect income smoothing via the accrual model. The following models are considered the most frequently employed in the literature:

(1) Healy's (1985) model.

General formula is as follows:

$$DA_{i,t} = \frac{TA_{i,t}}{A_{i,t}} \tag{1}$$

where $DA_{i,t}$ is discretionary accrual for firm *i* in year *t*; $TA_{i,t}$ is total accruals for firm *i* in year *t*; and $A_{i,t}$ is total assets for firm *i* in year *t*.



303

JAEE 5.3

304

(2) DeAngelo's (1986) model.

General formula is as follows:

$$DA_{i,t} = \frac{(TA_{i,t} - TA_{i,t-1})}{A_{i,t}}$$
(2)

(3) Jones' (1991) model.

First, estimate the time series regression for the sample as follows:

$$\frac{TA_{i,t}}{A_{i,t}} = \beta_0 \left(\frac{1}{A_{i,t-1}}\right) + \beta_1 \left(\frac{\Delta REV_{i,t}}{A_{i,t-1}}\right) + \beta_2 \left(\frac{PPE_{i,t}}{A_{i,t-1}}\right) + \varepsilon_{i,t} \tag{3}$$

where $TA_{i,t}$ is total accruals for firm *i* in year *t*; $A_{i,t-1}$ is total assets for the previous period; $\Delta REV_{i,t}$ is change in revenues for firm *i* in period *t*; and $PPE_{i,t}$ is gross plant property and equipment for firm *i* in period *t*.

Then, estimate discretionary accruals as stated as follows:

$$DA_{i,t} = \frac{TA_{i,t}}{A_{i,t}} - \left(b_0 \left(\frac{1}{A_{i,t-1}}\right) + b_{1i} \left(\frac{\Delta REV_{i,t}}{A_{i,t-1}}\right) + b_{2i} \left(\frac{PPE_{i,t}}{A_{i,t-1}}\right)\right)$$
(4)

(4) The cross-sectional Jones model by Defond and Jiambalvo (1994):

In order to overcome the survivorship bias generated from applying a time series approach in Jones' model, Defond and Jiambalvo (1994) adopt a cross-sectional version of Equation (3) using portfolios of firms based on industries.

(5) Modified Jones (1991) model:

Dechow *et al.* (1995) provide a modified version of Jones' model. They adjust the change in revenues by the change in receivables (debtors) during the event period in order to capture the manipulation of sales since the original version of Jones model assume that sales will not be manipulated by the company, this assumption contradict with Kaplan (1985) argument that sales can be used as tool of manipulating income.

3.2 GCC studies

The earlier discussion in Section 3.1 addresses well-known studies in this field in a general fashion. In order to address what was done in this field at the region of the study, this section will present various previous studies related to the GCC in this field in addition to other related studies discussed in Section 1.

Darayesh *et al.*'s (2010) study addresses the income enhancement practice in GCC, no attempt has been made to obtain different results for each GCC country or per sector. The frequency of the second digit for income numbers was used as an indicator for income enhancement engagement based on the assumption that the second digit distribution will reveal income engagement. Overall, results were mixed and the overall statistical results were insignificant as some digits are significant and others show unexpected trends. The conclusion was that income enhancement is somewhat present. Marashdeh and Shrestha (2010) examine the extent of the market integration in the GCC markets; the results show that GCC markets are not fully integrated. For example Kuwait market and Saudi Arabia markets are moving in opposite directions; results also show opportunities for international investment diversification. These support earlier

conclusions made by Abraham et al. (2001) when they analyzed three GCC markets, Evidence from namely, Bahrain, Kuwait and Saudi Arabia, and correlated the monthly market index with S&P 500 monthly market index return. The outcome of the study shows significant potential diversification opportunities based on low-negative correlation for Bahrain and Kuwait, and almost zero for Saudi Arabia with S&P market index.

The current study builds on what is done in the literature in this region and extends the contribution in terms of being the first to address income smoothing matters based on documented methodology. Further, more factors are included, such as audit quality and corporate governance, and further discussion about the additional two more factors can be found in Section 4. Moreover, other studies in the region will be mentioned in subsequent sections as the need arise.

4. Detecting income smoothing

Dechow et al. (1995) present real concerns about the ability of the Jones model, as the most famous and applicable accrual model, to separate accruals into discretionary and non-discretionary components. Other models are subject to the same criticism, as Jones' model is considered an advanced version of the other models. This justifies the existence of several versions of the lones model. On the other hand, many studies discussed in the previous section adopted the coefficient of variations method. In this study, the researcher will adopt the coefficient of variations approach, as it is more applicable to the study data set, taking into account the difficulties of collecting data from several countries and the current lack of a comprehensive data for this region. Further, the potential misspecification may create noise and harm the originality of the outcome. Finally, adopting advanced models across different countries requires more preparation and more studies to assess the applicability of adopting the same model for different countries in the same study[7].

As the main aim of this study is to assess the impact of income smoothing on earnings quality in the GCC capital markets, the proxy of income smoothing will be determined based on the coefficient of variation approach. Thus, the methodology of the study in this regard will follow the coefficient of variation method developed by Eckel (1981), which has been adopted in many studies, as discussed in the previous section. For example, Ashari et al. (1994), Albrecht and Richardson (1990), Tseng and Wen Lai (2007), Michelson et al. (2000) and Avoib and Nooriha (2009) adopt the coefficient of variation approach to detect income smoothing. This methodology measures income smoothing by accumulating the impact of a potential smoothing variable and considering it over time. The procedure is as follows:

Compute the coefficient of variation for the change in income and the change in sales, and then classify the company as smoother if Equation (5) is satisfied:

$$\frac{CV_{\Delta I}}{CV_{\Delta s}} < 1 \tag{5}$$

where ΔI is the one period change in income; ΔIS is one period change in sale; and CV is the coefficient of variation. CV is computed by dividing the standard deviation over the mean $CV = SD/\overline{X}$ (Berenson *et al.*, 2006, p. 86).

In order to extend the potential contribution of the study, two more factors will be added to this study[8]. First, is the audit quality. Teoh and Wong (1993), Balsam et al. (2003) and Krishnan et al. (2005), among others, document a relationship between audit



305

GCC markets

quality and earnings quality. Thus, in this study, the audit quality will be included as an additional factor to help understand the potential impact of income smoothing on earnings quality. Audit quality will be measured based on brand name as a proxy for this factor, data of the study split into two groups: the first is companies audited by any of the Big 4 auditing firms: Deloitte Touche Tohmatsu, Ernst & Young, KPMG or PwC. This group will expect to represent higher auditing quality, while the group audited by companies not in the Big 4 list represents relatively lower auditing quality. The Big 4 companies assume to be greater in size than non-Big 4 companies; this approach is adopted in other studies, such as Chen et al. (2010), that adopt similar a proxy based on another emerging market, which is the Chinese market. However, there is one difference that the current study limits the brand name to the international Big 4 companies for all GCC capital markets in order to minimize any noise due to the variations among GCC markets and since Big 4 classification is widely accepted in the region and abroad. Avoib and Nooriha (2009) used a brand name as a proxy for auditing quality and they apply the study to another emerging market, which is the Malavsian market.

The second factor is corporate governance. Marra *et al.* (2011) and Wang (2006), among others, argue that corporate governance is a potential factor that has an impact on earnings quality as a consequence of its role in public disclosure. Eventually, this will have an effect on earnings disclosure and quality. Thus, this study will add this new factor in order to achieve better understanding of the potential impact of the income smoothing on earnings quality. Two groups of companies are created based on the structure of the board of directors. The first group is for companies owned largely by the government, royal families or dominated by other companies with the same features; while the second group includes companies that have more diverse ownership among individuals including members of well-known families in the GCC markets or other corporate companies as institutional investors. Saidi (2011) argues that, corporate governance issue has been improved in the GCC region, as almost all GCC countries have a corporate governance code or guidelines in place. For example, Oman issued corporate governance standards for listed companies in 2002. However, he added, the next challenge is the effective implementation in the region.

5. Hypothesis of the study

As explained in the previous sections, there are two points of view about the impact of income smoothing on earnings quality. Managers could exercise discretion on accruals either to enhance the quality of earnings or to manipulate earnings opportunistically in order to pursue their own interests. The conflict between these two views is related to the way in which managers exercise their discretion over accruals. The question here is whether managerial discretion over specific components of accruals reduces or enhances the quality of earnings. The assumption in this study is that on average, managers use their discretion within GAAP to increase the ability of earnings to reflect company performance. The approach adopted here is based on a large sample. The other assumption of this study, based on the efficient market hypothesis (EMH), is that price is a valid benchmark for company performance; therefore any information in earnings or cash flow is revealed in the stock price. The main hypothesis of the study stated in the alternative form, is the following:

H1. On average, income smoothing is expected to enhance accounting earnings'



JAEE

To test this hypothesis, a simple linear regression will be run independently for Evidence from smoothing vs non-smoothing companies. The independent variable in the model is earnings per share deflated by the price at the beginning of the period to reduce the heteroskedasticity problem (White, 1980), while the dependent variable is stock returns. It is worth mentioning that various studies across accounting literature have used earnings levels deflated by price at the beginning of the period as an explanatory variable in the return model. Therefore, this methodology is well-documented in many studies. Kothari (2001) refers to two 1992 papers: Kothari (1992) and Ohlson and Shroff (1992), which provide a rationale for using earnings levels in returns – earnings regressions and states "[...] In recent years, researchers estimating a return-earnings regression frequently use earnings-deflated-by-price variable to explain stock returns [...]" (p. 129). In further detail: Kothari (1992) shows that earnings level deflated by price is better than using earnings-change-deflated-by-price. Further, earnings level deflated by price at the beginning of the period is better than earnings level deflated by lagged earnings variables. While, Ohlson and Shroff (1992) state that earnings-level variable should be the natural start point in explaining returns. On a related note, Dechow's (1994) study is very important in market-based accounting research; at it used earnings level deflated by price at the beginning of the period as an explanatory variable for returns. This will not rule out the opposing view which prefers to include earnings changes as sole or additional to earnings levels. Based on kind feedback from anonymous reviewers, a new model will be added, Model 7, which is based on earnings changes' interaction with a dummy variable of income smoothing, based on the same deflator:

$$R_{i,t} = \alpha + \beta E_{i,t} + \varepsilon_{i,t} \tag{6}$$

where $R_{i,t}$ is the stock return for company *i* at time *t*, $E_{i,t}$ is the EPS for company *i* at time *t* deflated by the price at the beginning of the period.

Equation (6) will run again by including dummy variables to account for the potential impact of audit quality and corporate governance on the association of earnings with stock returns, as indicated in Equations (6.1) and (6.2), respectively:

$$R_{i,t} = \alpha + \beta_1 \times E_{i,t} + \beta_2 A Q_{i,t} + \beta_3 E_{i,t} \times A Q_{i,t} + \varepsilon_{i,t}$$

$$(6.1)$$

where $R_{i,t}$ is the stock return for company *i* at time *t*; $E_{i,t}$ is the EPS for company *i* at time t deflated by the price at the beginning of the period; AQ_{it} is the audit quality dummy variable for company *i* at time *t* where $AQ_{i,t} = 1$ if company audited by any of the Big 4 auditing firms and zero otherwise; and $E_{i,t} \times AQ_{i,t}$ is the interaction between the audit quality dummy variable and EPS for company *i* at time *t*.

The criterion for testing this model is based on whether β_3 is significantly larger than zero, where β_3 measures the impact of auditor quality on the association between earnings and returns. A positive sign indicates a positive impact of audit quality on the association of earnings with stock returns for company i at time t and negative impact indicates the opposite:

$$R_{i,t} = \alpha + \beta_1 \times E_{i,t} + \beta_2 C G_{i,t} + \beta_3 E_{i,t} \times C G_{i,t} + \varepsilon_{i,t}$$

$$(6.2)$$

where $R_{i,t}$ is the stock return for company *i* at time *t*; $E_{i,t}$ is the EPS for company *i* at time t deflated by the price at the beginning of the period; $CG_{i,t}$ is the corporate governance dummy variable for company i at time t where $CG_{it} = 1$ if company ownership dominated by government and/or royal families, and zero when ownership



307

GCC markets

divided among institutional and individual owners; and $E_{i,t} \times CG_{i,t}$ is the interaction between the corporate governance dummy variable and EPS for company *i* at time *t*.

The criterion for testing this model is based on whether β_3 is significantly larger than zero, where β_3 measures the impacts of corporate governance on the association between earnings and returns. A positive sign indicates positive impact of corporate governance on the association of earnings with stock returns for company *i* at time *t* and negative impact indicates the opposite.

As indicated earlier, in order to strengthen the outcome of the study, an additional approach is to test the hypothesis based on earnings changes instead of earnings level as stated in Equation (6)[9]. Equation (7) lists model based on earnings changes with two explanatory variables; earnings changes and the interaction between a dummy variable of income smoothing and earnings changes[10]:

$$R_{i,t} = \alpha + \beta_1 \times DE_{i,t} + \beta_2 DUM_{i,t} + \beta_3 DE_{i,t} \times DUM_{i,t} + \varepsilon_{i,t}$$
(7)

where $R_{i,t}$ is the stock return for company *i* at time *t*; $DE_{i,t}$ is the change of EPS for company *i* at time *t* deflated by the price at the beginning of the period; $DUM_{i,t}$ is the income smoothing dummy variable for company *i* at time *t* where $DUM_{i,t} = 1$ if company classified as smoother based on Equation (5) and zero otherwise; and $DE_{i,t} \times DUM_{i,t}$ is the interaction between then income smoothing dummy variable and change of EPS for company *i* at time *t*.

The criterion for testing this model is based on whether β_3 is significantly larger than zero.

6. Sample of the study and descriptive statistics

Data of the study derived from the industrial sector of the GCC markets; titles of sectors vary among GCC countries[11]. Ultimately, all companies in the study are classified as industrial sector; companies included in the study must have sufficient financial data for a least five consecutive years, along with stock returns, in order to generate reliable estimates. This condition generates an initial sample size of 55 companies representing 72 percent of industrial companies in the GCC markets. As a breakdown of figures, companies in Kuwait included in the initial sample were 17 out of 24, in Oman ten out of 15, in Qatar five out of six, in Saudi Arabia 14 out of 21, in UAE nine out of ten and in Bahrain no company was classified in the industrial sector. In summary, 55 companies in the GCC market extracted from 76 companies in the industrial sector, based on this high percentage this initial sample, could be regarded as representing the population of listed industrial companies in these emerging markets.

This study is based on data collected from GCC capital markets for ten years from 1999 to 2008. The initial number of company-year observations before excluding outliers was 447[12]. The use of this time period kept the financial crisis from having a significant impact on the results[13]. The data were limited to the industrial sector, the GCC countries have sought to adopt similar policies. However, due to differences in several factors, industry classifications are not harmonized in titles across the capital markets of the GCC. Further, countries like Bahrain have been significantly dominated by the financial sector and there were no data as industrial sector in this market, so Bahrain market was dropped from the study due to the lack of representative data. The United Arab Emirates (UAE) had two markets, one in Dubai and the other in Abu Dhabi. For the sake of standardization, these two markets were combined as the UAE market; observations for this market came mainly from Abu Dhabi market as Dubai



JAEE 5.3 market. like Bahrain market, does not have representative data for the industrial sector. Evidence from However, results were unexpected, when computing the smoothing index the outcome shows that all UAE observations were classified as non-smoother companies thus the entire company-year observations for this market were dropped this cause losing 70 company-year observations derived from nine companies related to this market. The reason of excluding UAE data since there was no company engaged in income smoothing within the market based on the results[14]. Thus, there is no chance for any comparison so UAE observations will be excluded from the analysis, as there are only non-smoothing companies in UAE. Moreover, the outliers were excluded based on the main variables per country[15]. Thus, the total number of lost observations is 122 company-year observations and that made the final sample as measured in companyvear observations declined to 325. Table I describes the cause of this decline in detail.

Table II shows the distribution of annual observations among different countries. Final sample size represents 64 percent of industrial companies for the remaining four GCC countries[16]. Table III presents the distribution of company-year observations among different countries. Tables II and III show that most of the observations came from the Saudi Arabia and Kuwait markets. All companies are from the industrial sector of GCC markets and combined at country level as industrial companies, in order to ensure consistency throughout the paper.

The smoothing index was computed according to Equation (5) and classified the companies as smoothing or non-smoothing. Initial results before excluding outliers show that 140 observations out of the sample size, 447 observations, were classified as income smoother. Thus, distribution of smoothing vs non-smoothing companies

Cause	Lost observations: company-year	
Excluding UAE observations	70	
Outliers in SA	11	
Outliers in KW	31	Table I.
Outliers in OM	10	Causes of decline in
Outliers in QA	0	company-year
Total	122	observations

Country	Frequency	%	Cumulative frequency	Cumulative (%)	Table II.
KW: Kuwait OM: Oman	14 10	33.3 23.8	14 24	33.3 57 1	Distribution of data according to number of
QA: Qatar SA: Saudi Arabia	5 13	11.9 31	29 42	69 100	companies per country

Country	Frequency	%	Cumulative frequency	Cumulative (%)	Table III. Distribution of data
KW: Kuwait	98	30.15	98	30.15	according to
OM: Oman	84	25.85	182	56.00	company-year
QA: Qatar	32	9.85	214	65.85	observations
SA: Saudi Arabia	111	34.15	325	100.00	per country



309

JAEE
 5,3
 reveals that income smoothing is practiced by GCC companies. About one-third of GCC companies are classified as income smoothing companies. In order to shed light on smoothing behavior among countries in the GCC, the income smoother behavior was examined for each country. Initial results show that smoothing companies equate to 31.3 percent of the total before excluding outliers. After excluding UAE due to earlier explanation, the percentage of smoother companies has become 37.14 percent. This percentage is calculated by dividing the 140 company-year observations classified as smoother by the total number of the observations (447), subtracting the 70 UAE company-year observations:

$$\frac{140}{447 - 70} \times 100\% = 37.14\%.$$

The percentage of smoothing company-year observations varies across GCC courtiers. For Qatar and Saudi companies, this figure was 59.38 and 51.64 percent, respectively while it was just 28.7 and 24.03 percent for Kuwait and Oman companies, respectively. The results in Table IV show that the percentage of smoothers, after excluding UAE observations and outliers, is 41.85 percent, while in Table V results show that the percentage of observations classified as smoother are 59.38, 54.95, 30.95 and 30.61 percent for Qatar, Saudi Arabia, Oman and Qatar, respectively[17].

7. Hypothesis testing and inferential statistics

Earnings quality is measured based on the ability of earnings per share deflated by the price at the beginning of the period to explain variations in stock returns for the same period. Following the efficient market hypothesis, stock returns can be used as a base for earnings quality, as the EMH states that prices reflect any relevant information about the company in a timely fashion. Higher coefficient of determination (adjusted R^2) in stock returns-earnings model reflects higher earnings' quality. Marketbased accounting research studies adopt this assumption. In order to measure the

Table IV.	Smoothing index	Frequency	%	Cumulative frequen	су	Cumulative (%)
Smoothing index for all company-year observations	Smoothing Non-smoothing	136 189	41.85 58.15	136 325		41.85 100.00
	Index	KW	OM	Country QA	SA	Total
	Smoother % Row (%) Col (%)	30 9.23 22.06 30.61	26 8.00 19.12 30.95	19 5.85 13.97 59.38	61 18.77 44.85 54.95	136 41.85
Table V. Smoothing index for each country, company-year observations	Non-smoother % Row (%) Col (%) Total %	68 20.92 35.98 69.39 98 30.15	58 17.85 30.69 69.05 84 25.85	$ \begin{array}{c} 13\\ 4.00\\ 6.88\\ 40.63\\ 32\\ 9.85 \end{array} $	50 15.38 26.46 45.05 111 34.15	189 58.15 325 100.00



impact of income smoothing on earnings quality, the simple linear model indicated in Evidence from Equation (6) was run independently for all data for smoothing and non-smoothing companies. Table VI presents results for the regression model for smoothing and then non-smoothing companies.

Results in Table VI suggest that, on average, income smoothing as it is practiced in GCC industrial companies does not reduce the ability of earnings to reflect company performance, so there is no negative impact on earnings quality by engaging in income smoothing in companies within the GCC markets. However, results suggest that the gap between the quality of earnings in the smoothing companies vs the non-smoothing subsample is relatively very small, with a relative advantage accorded to the smoother companies. Vuong (1989) developed a technique to compare two regression models and to test whether the relative differences are significant or not. When one applies the Vuong technique of the two models in Table VI, the *p*-value of the computed Z value only equals 12 percent[18]. Assuming that the confidence level adopted in the study is 90 percent, this suggests that there is no significant difference between the ability of earnings to explain variations in stock returns between smoothing or non-smoothing companies, as the Zvalue of the differences is not significant on a 90 percent confidence level.

For a further investigation the test of the impact of income smoothing on the earnings quality repeated for each country. Results, presented in Table VII, show clear variations among the four countries in the GCC included in the study. Saudi Arabia's observations provide evidence that involvement in income smoothing causes a decline in earnings quality, as the explained variations in stock returns by earnings for non-smoothing companies are higher than those for smoothing companies, while the opposite was noticed for other countries. The mixed results show differences among different countries within the GCC[19].

Thus, the overall conclusion is that the hypothesis of the study is accepted for three out of four countries. On the aggregate level, however, the insignificant results for the pooled sample are driven by the observations on Saudi Arabia companies, therefore, despite the relative improvement of income smoothing on earnings quality, the overall impact is not significant. This suggests that the hypothesis is accepted for three countries and rejected for one. The initial conclusion was that the alternative hypothesis can be accepted for group 1 (including Kuwait, Oman and Qatar), and rejected for group 2 (only Saudi Arabia). The researcher will apply an alternative model based on earnings changes as stated in Model 7 for the pooled data, in order to assess the impact of income smoothing on aggregate level based on earnings changes. Related results will be discussed later in this section and results will be presented in Table XII.

Annual	E smoother companies	E non-smoother companies	
Coefficient (t-statistic) Number of observations Adj. $(R^2)\%$ R_{Es}^2/R_{Ens}^2 b-value/Vuong test	3.08 (4.71) 136 13.59	2.12 (5.34) 189 12.78 0.12	Table VI. A comparison of the association of earnings with stock
Note: The table gives results t	from simple regression of stock retu	rns on earnings per share deflated	smoothing and

Note: The table gives results from simple regression of stock returns on earnings per share deflated by the market price at the beginning of the period for smoothing vs non-smoothing companies. $R_{i,t} = \alpha + \beta E_{i,t} + \varepsilon_{i,t}$



GCC markets

311

non-smoothing

companies

JAEE 5.3	Country/results	KW: Kuwait	OM: Oman	QA: Qatar	SA: Saudi Arabia		
0,0	Panel A. Smoother comban	ies					
	Coefficient (<i>t</i> -statistic)	2.78 (5.41)	4.33 (3.31)	10.66 (2.33)	1.99 (1.30)		
	Number of observations	30	26	19	61		
219	Adj. (<i>R</i> ²)%	49.37	28.42	19.70	1.15		
Table VII. A comparison of the association of	Panel B: Non-smoother com Coefficient (t-statistic) Number of observations Adj. (R ²)%	panies 2.93 (5.47) 68 30.11	1.37 (2.32) 58 07.12	4.02 (1.72) 13 <i>13.94</i>	6.23 (1.82) 50 17.94		
earnings with stock returns between smoothing and non-smoothing companies for each country	Panel C: Adj. (R^2) for smoother companies vs non-smoother for each country R_{Es}^2/R_{Ens}^2 0.06Note: The table gives results from simple regression of stock returns on earnings per share deflated by the market price at the beginning of the period for smoothing vs non-smoothing companies for each country. $R_{i,t} = \alpha + \beta E_{i,t} + \epsilon_{i,t}$						

Regarding Saudi Arabia's opposing results (as stated in Table VII), the Saudi market is the most recent of the four markets addressed in Table VII. Moreover, Camilla Hall, in a Financial Times article published on January 23, 2012, mentioned that Saudi Arabia is working toward opening its borders to foreign investment (Hall, 2012). This fact may change the earning environment in this market as, during the period of the study, foreign investment is not allowed to enter directly the Saudi market, which differentiates it from other GCCs as other countries are more open to foreign investment to various extents. Thus, one of the potential explanations for the negative impact of income smoothing on earnings' quality as an indicator of company performance in the Saudi market, may be due to the strong regulatory enforcement. This led to restricting accountants from signaling to the market and limiting income smoothing to communicate the positive aspect of earnings. In the same article, Camilla Hall, mentioned that Saudi Arabia had a tough stance in terms of regulations; so, applying this fact to the study it seems that the market is not welcoming any flexibility in terms of applying income smoothing tools by accountants. Thus, such behavior, where it exists, had a negative impact on the ability of earnings to reflect company performance. This may explain the differences in Saudi Arabia market compared with other markets in this regard. On the other hand, the Kuwait market is the oldest market in the GCC, followed by Oman, Qatar and, finally, Saudi Arabia. It seems that practice shaped over time, and it may improve in the Saudi market if this study is replicated after a few years.

Further analyses lead to conclude that income smoothing as action in this study has a relatively positive impact, in general, on earnings. Therefore, the results of this study are in line with the group of studies mentioned in footnote 1 that state the positive impact of smoothing. As a descriptive measure, the EPS mean of the smoother group is more stable compared with that of the non-smoother group as it has less standard deviation and narrower range. Furthermore, the median for smoother is closer to the mean compared with the distance between mean and median of non-smoother. This enhances the stability conclusion about the mean of smoother. In figures, the EPS mean, median, standard deviation and range for smoother and (non-smoother) were 1.19 (0.86), 0.67 (0.23), 1.60 (1.63) and 8.74 (12.95), respectively. Moreover, as it was noted



in Section 6 footnote 14, outliers excluded were mostly non-smoother. This exclusion Evidence from enhances the non-smoother model and narrows the gap with the smoother. The exclusion of outliers is a normal action done in any research to maintain objectivity and avoid results being dominated by outliers.

A further sensitivity analysis was conducted to check the robustness of the outcome from the earnings-returns relationship. The model was conducted for each year of the study, data of the study was divided among period of the study as the number of observations and (percentage) for years 2000 to 2008 were 33 (10.15), 31 (9.54), 32 (9.85), 31 (9.54), 35 (10.77), 39 (12.00), 42 (12.73), 41 (12.62) and 41 (12.62), respectively. This distribution shows that no single year dominates the sample, and χ^2 computed a very small value and equal to 0.131, while the critical χ^2 value at eight degree of freedom is 20; therefore, it is obvious that this computed value is not significant[20]. Moreover, in order to keep more observations, the smoother was added as a dummy to the model and the model ran for the whole study. The coefficient of smoother dummy (0 non-smoother and 1 smoother) was positive which confirm the overall advantage of smoother sample over non-smoother whereas the significant level was not high enough to secure sufficient evidence and this analysis was repeated by year and again the trend stay the same in most years.

As indicated in Section 4, two more factors are added to this study. The first is audit quality, which is included to explore the audit quality level in the four GCC countries and to measure the impact of this variable on the association between earnings and returns. While the second factor is corporate governance, this is included for the same reason as the audit quality variable. Tables VIII and IX present results of Models 6.1 and 6.2, respectively, which are related to audit quality and corporate governance, respectively. In Table VIII, results show sign of β_3 which reveals that the impact of auditor quality on the association between earnings and stock returns is positive and equal to 0.87. This positive sign of the interaction between the audit quality dummy variable and earnings indicates that the impact of audit quality on income smoothing is positive. This trend coincides with the expectation presented in Section 4, and with

Annual	Intercept	$\beta_1 E_{i,t}$	$\beta_1 A Q_{i,t}$	$\beta_3 E_{i,t} \times AQ_{i,t}$
Coefficient (t-statistic)	-0.07 (-1.05)	1.89 (3.46)	-0.03 (-0.43)	0.87 (1.23)
Adj. (R^2) %		. ,	12.83	

Note: The table displays results from regression of stock returns on earnings per share deflated by the market price at the beginning of the period, the dummy variable for audit quality and interaction between audit quality and earnings as presented in Equation (6.1)

Annual	Intercept	$\beta_1 E_{i,t}$	$\beta_2 CG_{i,t}$	$\beta_3 E_{i,t} \times CG_{i,t}$
Coefficient (t-statistic)	-0.07 (-1.58)	2.09 (4.91)	-0.07 (-0.88)	0.95 (1.28) In
Adj. (R^2) %			12.81	

Note: The table gives results from regression of stock returns on earnings per share deflated by the market price at the beginning of the period, the dummy variable for corporate governance and interaction between corporate governance and earnings as presented in Equation (6.2)



GCC markets

313

Table VIII. Impact of audit quality on association between stock returns and earnings

Table IX. pact of corporate governance on association between stock returns and earnings the trend of other studies listed in Section 4 also which were conducted on other emerging markets, such as Avoib and Nooriha (2009) and Chen *et al.* (2010). However, the coefficient value, 0.87, is not significant as the *p*-value reached 22 percent, which generated a computed confidence level of only 78 percent, which is less than the accepted confidence level of the study (90 percent). The same observation is noted for the corporate governance model; results presented in Table IX reveal that the sign is positive and the amount of β_3 which measures the impact of corporate governance on the association between earnings and stock returns equals 0.95, with a p-value reaching 20 percent. Therefore this indicates a better association between earnings and stock returns for companies dominated by government, but this trend is not significant.

Further analysis occurs to present more information concerning the impact of the two additional variables (audit quality and corporate governance) on the association between stock returns and earnings by splitting the data based on each factor alone. when the model as presented in Equation (6) runs for a subsample of each factor. Goodness of fit for 99 company-year observations audited by non-Big 4 auditors equals 11.4 percent, while it reached 13.7 percent for 226 company-year observations audited by one of the Big 4 auditors. Goodness of fit for 207 company-year observations governed by private sector equals 10.4 percent, while it reached 16.4 percent for 118 company-year observations governed by government and royal families. Both results coincide with results presented in Tables VII and VIII based on Equations (6.1) and (6.2), respectively. However, since results in Tables VIII and IX are not significant and the results mentioned in this paragraph have a descriptive nature, further analysis in future studies should be conducted before reaching a final conclusion regarding this matter. The study underlines the need for further research in the future.

The inclusion of two additional variables, audit quality and corporate governance, is extended to provide an example of each variable in the four GCC markets. Tables X and XI display further information about audit quality and corporate governance, respectively. Results in Table X show that almost 70 percent of company auditors are one of the Big 4 auditor firms or their partners, while the other 30 percent are mainly local auditors without connection to any of the Big 4.

Distributions of audit quality among the four countries show the same trend with various extents, majority of company-year observations in each country are related to one of the Big 4 auditor companies, as 53, 68, 76 and 100 percent for Kuwait, Oman, Saudi Arabia and Qatar, respectively. The 100 percent in Qatar for one of the Big 4 auditors firms may be due to the fact that Qatar investment funds undertook several

	KW	OM	Country QA	SA	Total
Big 4 auditors (1)	52 16.00	57 17 54	32	85 26 15	226
/0 Row (%) Col (%)	23.01 53.06	25.22 67.86	14.16 100.00	20.13 37.61 76.58	09.04
Local auditor (0)	46 14 15	27	0	26 8 00	99 20.46
^{/0} Row (%)	46.46	27.27 29.14	0.00	26.26 22.42	30.40
Total	40.94 98 30.15	32.14 84 25.85	32 9.85	23.42 111 34.15	325 100.00
	Big 4 auditors (1) % Row (%) Col (%) Local auditor (0) % Row (%) Col (%) Total %	KWBig 4 auditors (1) 52 %16.00Row (%)23.01Col (%) 53.06 Local auditor (0) 46 %14.15Row (%) 46.46 Col (%) 46.94 Total 98 % 30.15	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $



JAEE

huge investments in Europe and USA. Thus, in order to comply with the requirements Evidence from of all these advanced markets you have to maintain the highest level of audit quality, which may explain this exceptional percentage in Qatar for the high audit quality. On the aggregate level audit quality has a positive impact on earnings quality and this coincides with previous results, such as Teoh and Wong (1993) and Balsam et al. (2003). Moreover, as presented earlier, the goodness of fit of the model computed for two subsamples based on audit quality variable, coefficient of determination for non-Big 4 models with 99 observations was 11.38 percent, while it was 13.72 percent with 226 observations for the Big 4 model. This result goes in the same line of results in other studies that higher audit quality enhances earnings quality, like Balsam et al. (2003) who reach the same trend based on ERC; therefore, this result provides similar evidence in this region[21]. On the other hand, Balsam et al.'s (2003) study argues that more audit quality leads to less discretionary accruals, so they implicitly took the negative side about income smoothing by concluding that more income smoothing will lead to less earnings quality as it is measured by less ERC. In other words, it will reduce the quality of earnings. In this portion, this study did not find similar trends as, from the 226 observations classified in the Big 4 category, 97 were classified as smoother (43 percent), while 129 were classified as non-smoother (57 percent). On the other hand, 99 observations were classified as non-Big 4 companies distributed in observations (percentages) as 39 (39.4 percent) and 60 (60.6 percent) for smoother and non-smoother, respectively. Further, the computed χ^2 value for this contingency table was only 0.35 and this small value is not significant as the cut-off χ^2 at 90 percent confidence level equal 6.63. To sum up, audit quality leads to higher earnings quality and there is no evidence that high auditor quality leads to less smoothing in the region. This, again, is in the line with the group of studies that state that income smoothing could enhance earnings ability to reflect company performance.

The other, second, factor is the corporate governance. Sbeiti (2008) states that specialized banks in GCC are owned by government and commercial banks are owned by family owned with modest government participation. In the same direction, Al-Shammari et al. (2008) report that the institutional ownership in GCC only 0.20. This leads to a general impression that the corporate governance structure in the GCC countries is that most companies are dominated mainly by the government or by wealthy families. However, this impression, based on those aforementioned studies among other studies, reaches this conclusion based mainly on the financial sector, while this study explores the industrial sector. Thus, it is expected that the ownership

	17117	014	Country	<u> </u>	T + 1
Index	KW	OM	QA	SA	Total
Government dominated ownership (0)	8	40	27	43	118
%	2.46	12.31	8.31	13.23	36.31
Row (%)	6.78	33.90	22.88	36.44	
Col (%)	8.16	47.62	84.38	38.74	
Private dominated ownership (1)	90	44	5	68	207
%	27.69	13.54	1.54	20.92	63.69
Row (%)	43.48	21.26	2.42	32.85	
Col (%)	91.84	52.38	15.63	61.26	
Total	98	84	32	111	325
%	30.15	25.85	9.85	34.15	100.00



observations

GCC markets

JAEE structure in firms operating in the GCC markets will be dominated by government or family business. This matter was investigated by assessing the corporate governance of GCC firms analyzed in this study. All companies are divided into two categories: the first includes companies dominated heavily by government units or royal family or by other companies that having the same features; and the second comprises companies that have either commercial institutional owners or individuals. This ownership structure in the second group is more diverse than the first. There are 316 vague cases that have major shareholders from both sides at the same time, in this case the weight of each category was computed and the higher weight category was considered. This matter was not a black vs white classification, so there may be some noise. Results, as presented in Table XI, show that 118 company-year observations represent 36.31 percent was classified as belong to companies that denominated by governmental units or royal family figures while 207 companies stand for 63.69 percent classified as companies with relatively more diversifications in the ownerships. These results indicate more liberal policy and less than expected government involvement, in general, in the industrial sector, However, this trend is not harmonized among the four GCC countries; in the same table the distribution of the corporate governance for each country is presented as well. Results show clear variations among the four GCC countries with different weight that not necessarily goes in the same trend as indicated by the aggregate amounts. Kuwait and Qatar are two extremes as the vast majority of Kuwaiti companies, 91.8 percent, having diversified ownership structure while the contrary in Qatar that show the vast majority of Qatar companies, 84.4 percent, are dominated by the government. In Oman, results show almost equal division among the two groups while the relative majority in Saudi Arabia could be classified as having diversification in the ownership as 61.3 percent of company-year observations were classified in this category. The detailed results show different indicators, as in three out of four countries there were influential impact of the company-year observations that classified as dominated by the government or royal families, 38.7, 47.6 and 84.4 percent for Saudi Arabia, Oman and Qatar, respectively. While Kuwait had only 8.2 percent of company-year observations, they were classified as government dominated observations. Such results could be due to the fact that Kuwaiti market is the oldest market in the region despite the fact there is no code vet for corporate governance as indicated in Saidi (2011) study and in other studies; Kuwait took significant steps in the proper implementation and building more institutional controlling practices that help in diversifying ownership structure ahead of other countries in the region, this interesting outcome in this study coincide with results in Shammari et al. (2008) when it reports Kuwait at the top of all the GCC countries in terms of institutional ownership based on the general GCC sample that covers various sectors. Result, as presented earlier, favors government dominated companies due to the fact that market may fixate to this type of owners due to their influence but this point require more investigation in more depth in a separate study that consider more sectors, more data and control for other factors in order to reach solid conclusion. In addition, the distribution of the companies in terms of corporate governance and classification of smoother and non-smoother is done. Totally, 207 observations for the more diversity in ownership were distributed as 86 smoother with 41.55 percent and 121 as non-smoother with 58.45 percent of this category. The 118 observations classified as governmental dominated ownership were distributed as 50 smoother with 42.4 percent and 68 non-smoother with 57.6 percent. χ^2 computed value for the contingency table was just 0.02 and shows there is no relation in the smother distribution with the corporate governance and



both categories of corporate governance place almost the same trend in dealing with Evidence from GCC markets

Finally, the impact of income smoothing on earnings quality is re-tested based on Equation 7. This model will include the entire observations, utilizing changes in earnings and allowing for interaction between the income smoothing dummy variable and changes in earnings. If income smoothing improves earnings quality to measure company performance, then the coefficient of the interaction between income smoothing and changes in earnings is expected to be positive. The results presented in Table XII show that the β_3 sign is positive and this confirms the earlier results about the positive impact of income smoothing on earnings quality. Furthermore, this coefficient is significant (*p*-value equals 0.001). Thus, the changes in the earnings model confirmed the earlier results based on earnings level and provides clear evidence of the validity of the result.

Thus, based on results in Table XII, the overall conclusion is that income smoothing improves earnings quality and the model based on earnings changes provides stronger and clearer evidence, supporting the initial results based on earnings-level approach.

8. Conclusions and recommendations

This paper shows that the income smoothing practice in GCC countries exists, and has various effects. Further, it has shown that on average, income smoothing practice did not reduce the quality of earnings measured by the ability of earnings to explain variations in stock returns. On average, income smoothing improved the quality of earnings, but this advantage was not significant on the aggregate-level for the earnings-level model. On the country level, based on earnings-level model, there were variations. For instance, in Saudi Arabia, income smoothing reduced earnings quality, whereas it enhanced the quality of earnings in Kuwait, Oman and Qatar to different extents. The highest impact was in Oman, followed by Kuwait and Qatar. In order to obtain solid evidence, a new model was introduced based on earnings changes. Results for the whole sample based on this model revealed the positive impact that income smoothing has on improving earnings' ability to reflect company performance. This was made possible as the model provided significant evidence to confirm the positive impact of the interaction between earnings changes and income smoothing.

Overall, results of the study show that it correlates with results of other studies conducted in different markets which adopt a positive attitude toward the impact of income smoothing as presented in earlier sections. However, the study contradicts those adopting a negative attitude toward income smoothing.

This study explored income smoothing practice in GCC capital markets. This behavior is known and practiced in the area based on the results of four out of six markets that comprise the GCC capital markets. Further, the impact of income smoothing varies across GCC countries. The overall result revealed the positive impact that income smoothing has on earnings quality, with significant evidence based on the earnings

Annual	Intercept	$\beta_1 \times DE_{i,t}$	$\beta_1 DUM_{i,t}$	$\beta_1 DE_{i,t} \times DUM_{i,t}$	
Coefficient (p-value)	0.058 (0.133)	0.983 (0.001)	0.009 (0.876)	2.317 (0.001)	Table XII. The impact of income smoothing
Note: The table	gives results from reg	ression of stock return	ns on income smoothir	ng, changes in earnings	on the association

Note: The table gives results from regression of stock returns on income smoothing, changes in earnings per share deflated by the market price at the beginning of the period and the interaction between income smoothing and changes in earnings. $R_{i,t} = \alpha + \beta_1 \times DE_{i,t} + \beta_2 DUM_{i,t} + \beta_3 DE_{i,t} \times DUM_{i,t} + \varepsilon_{i,t}$



317

of earnings with

stock returns

changes model. Further, this study extends the topic by adding two more factors: audit quality and corporate governance. Most auditors belong to one of the large four auditor firms; this was recognized for all the GCC countries involved in this study. However, in corporate governance, the majority of companies have relatively more institutional private ownership with variation among different markets within the region. The two additional factors included in the model were based on dummy variables; both variables have a positive sign for their dummy variables in the model but only relative improvement.

This study contributes to the accounting literature by exploring this issue in the Middle East and GCC and provides various directions for future research in extending this research to other sectors and countries. The Saudi market requires further analysis in order to provide an answer for the different behavior compared to other markets within the GCC. The impact of the GCC economy movement during the period of the study may have an impact on the results; the GCC economy tripled in size to \$1.1 trillion from 2002 to 2008. Thus, extending the period to account for the financial crisis should be in the other direction for future studies.

The GCC consultative committee suggested changing the name of the GCC to Arab Gulf Union Council (Zawya, 2012). In order to enhance integration between markets of the GCC region, this study provides empirical evidence that there are variations among markets in the region and it encourages more analyses and further studies before proceeding to the union stage.

Acknowledgments

The author would like to thank the participants of the British Accounting Association North Group, who provided helpful comments during the conference held in Glasgow on August 30, 2010. The author would also like to thank his colleague Mostafa Jazayeri for arranging the research visit to MMU in summer 2012. Comments from the anonymous reviews had a significant impact on enhancing the paper and helped the researcher in adding two new dimensions to the original version and another approach in testing the hypothesis. The support and encouragement from the Global Office and School of Management at New York Institute of Technology are also gratefully acknowledged.

Notes

- 1. Lambert (1984), Trueman and Titman (1988), Subramanyam (1996), Sankar and Subramanyam (2001) and Hunt *et al.* (2000) adopt this perspective. Managers may use income smoothing to communicate private information about future earnings to the market. Thus, it is expected that income smoothing will improve earnings informativeness and quality.
- 2. Managers smooth earnings for their own benefit. Thus, it is expected that income smoothing will reduce the quality of earnings (Leuz *et al.*, 2003; Healy, 1985; Defond and Park, 1997).
- 3. The earliest study to use the term "income smoothing" explicitly was that of Hepworth (1953): "[...] Certainly the owners and creditors of an enterprise will feel more confident toward a corporate management which is able to report stable earnings than if considerable fluctuation of reported earnings exist [...] (p. 33)."
- 4. Eckel (1981) presents this type as management actions to control economic events.
- 5. None of the 44 countries were in the Middle East. Thus, the current study will not overlap with that of Cahan *et al.* (2008).



IAEE

- 6. Cahan et al. (2008) adopted the approach of Tucker and Zarowin (2006) in determining Evidence from earnings informativeness. The indicator for the earnings informativeness was the future GCC markets earnings response coefficient (ERC) which measures the association between current stock returns and future earnings. While the indicator for income smoothing was based on accrual approach by measuring the Spearman correlation between the cash from operations and total accruals over five years for each firm; a negative correlation was an indicator of income smoothing. The study avoided using the Jones model or any other accrual model to prevent misspecification since data divided among 44 countries.
- 7. There are studies that have taken the same direction and used simple models, rather than versions of Jone's model. This is because the number of observations per industry is limited or there is no variation in the industry; as in the current study, the examination based on these models will be unreliable (Wysocki, 2004). Marra et al. (2011) adopted this point of view when they looked at abnormal working capital accruals as a proxy for earnings management, since their data, based on Italian companies, have limited year/industry observations.
- 8. I would like to acknowledge the inclusion of the two factors based on the anonymous referee feedback and invaluable discussion.
- 9. As mentioned prior to presenting Equation (6), using the earnings level as an explanatory variable in the returns-earnings model is a well-documented approach and utilized in various studies. However, this will not rule out the opposing view, which prefers to include earnings changes as sole or additional to earnings levels. Thus, this model based on earnings changes' interaction with a dummy variable of income smoothing provides an alternative approach.
- 10. An additional year was collected in order to maintain the first year of the study. By doing this, the sample size is maintained for the new model, identical to Model 6, but based on earnings changes.
- 11. Industrial companies in the Saudi market are divided among industrial and petrochemical sector, cement sector while the agriculture sector dropped since this sector has its own characteristics and no other country in the gulf had a counter (similar) sector. For simplicity and due to the limited number of companies in each sector the two sectors combined to represent the industrial sector of Saudi Arabia market. Kuwait had one industrial sector that include both petrochemical and cement companies; the same policy adopted in Qatar. UAE had two markets, one located in Dubai and the other in Abu Dhabi. The Abu Dhabi market had a construction sector and industrial both combined together as industrial sector later Abu Dhabi market, while Dubai combined construction with real estate in the same sector. As for Bahrain there is no company classified as industrial company during the study period.
- 12. Data Hand Collection started in 1999 as earliest available data for the research, since variables need to be deflated with one year lag as explained in Section 3, data for 1999 secured variables for 2000 for a deflate but observations for 1999 needs deflated for 1998 which most of the data were not available. Thus, in the coming sections you will notice that the analysis done from 2000 to 2008 and observations are distributed nicely among these years.
- 13. This matter needs to be addressed in a separate paper by extending the period to include more years after the crisis once enough data are available.
- 14. UAE is considered as a leader in bringing foreign investment to the region, despite the fact that foreign investment attraction needs more effort as indicated in Shammari et al. (2008) study. In addition UAE has a unique situation in the region as it is the only state with two markets. Further, UAE liberal economic policy led to the high integration of the private and public sector. So, this may cause to limit discretion of accountant to convey any message via income smoothing. Thus, it leads to have all UAE observations classified as non-smoother. However, this matter requires further investigation as the current study is limited to the industrial sector, so when other sectors are included in future studies, the justification will be more mature.



319

- 15. The data were standardized in terms of currency by transferring all financial data to US dollars. Given the variation of the currency amounts among the GCC countries, it was preferable better to exclude outliers per country rather than for the entire data set. Outliers determined by exclusion the extreme values of stock returns, smoothing index and deflated earnings per share for each country. Company-year observations for Qatar were small and represent a small portion of the total. Further, there were not huge variations in the Qatar observations; thus, no observations from the Qatar market were deleted.
 - 16. This percentage will be regarded as the final percentage and is reported in the abstract, the earlier percentage (72 percent), listed on the previous page is an initial percentage before excluding UAE observations and before eliminating outlier observations as listed in Table II.
 - 17. It is obvious that percentage of smoother increased from 37.14 to 41.85 percent after excluding outliers except for Qatar as no observations were excluded. Thus, from this initial indicator most of outliers were from the non-smoother category. This shows that the smoother observations are steadier compared with the non-smoother category. More analysis will be conducted related to this point in Section 7.
 - 18. The Voung test assumed that the two non-tested models were developed based on the same sample size. In order to maintain this assumption, the non-smoother subsample was modified in selecting the same number of smoother size. Thus, since the number of company-year observations of the smoothers are 136 observations, three non-smoother subsamples with 136 observations were created randomly; the first subsample was created by selecting first 136 company-year observations from the non-smoother, the second by selecting the bottom 136 observations and the third was created starting from the middle, 94th observation, then select the 94th observation with 67 observations upwards and then another 68 downwards from the middle. Then Voung test runs independently three times to compare the smoother sample and three overlapped sub non-smoother samples that match the smoother in size; 136 observations each. *p*-Value of the *Z* score that created as output from Voung test, results show that goodness of fit for the smoother was relatively higher than that for the non-smoother. However, this superiority was insignificant as the higher significance level for *Z* score reached 87 percent; this cannot be considered as sufficient evidence, as the minimum significance level should be higher than 90 percent.
 - 19. Al-Shammari *et al.* (2008) present the fact that, despite strong ties between the GCC, there is diversity among GCC countries in terms of compliance with international accounting standards. Furthermore, Al-Hussaini *et al.* (2008) conclude that the extent of monitoring compliance with accounting standards varies between GCC countries.
 - 20. Computations of χ^2 done based on Keller (2005, p. 559).
 - 21. ERC, as pointed out and utilized in this study and other studies discussed earlier, could be a good alternative for the return-EPS model as it adopted in this study. However, the current methodology adopted in order to maintain the current sample size at this level and avoid losing more observations as more data will be needed data to determine unexpected earnings and unexpected return.
 - 22. This study explores the potential relationship between the distribution of auditor quality and corporate governance to open the door for further research to investigate this matter. From 118 observations classified as governmental dominated observations; 92 observations stand for 78 percent were having auditor as one of the Big 4; and 26 observations with 22 percent as auditor not from the Big 4 category. The 207 observations classified as more institutional ownership have the same trend but with less extent as from 207 observations 134 with 64.7 percent were smoother and 73 observations with 35.3 percent were non-smoother the compute χ^2 for this contingency table is 6.2, which is still not significant as critical value at 90 percent confidence level is 6.6 but it is closer and it shows that this topic should be explored in more depth to achieve better understanding of the GCC market.



JAEE

References

- Abraham, A., Seyyed, F.J. and Al-Elg, A. (2001), "Analysis of diversification benefits of investing in the emerging gulf equity markets", *Managerial Finance*, Vol. 27 No. 10, pp. 47-57.
- Albrecht, W.D. and Richardson, F.M. (1990), "Income smoothing by economy sector", *Journal of Business Finance and Accounting*, Vol. 17 No. 5, pp. 713-730.
- Al-Hussaini, A., Al-Shammari, B. and Al-Sultan, W. (2008), "Development of enforcement mechanisms following adoption of international accounting standards in the gulf co-operation council member states", *International Journal of Business Strategy*, Vol. 8 No. 3, pp. 50-71.
- Al-Shammari, B., Brown, P. and Tarca, A. (2008), "An investigation of compliance with international accounting standards by listed companies in the gulf co-operation council member states", *The International Journal of Accounting*, Vol. 43 No. 4, pp. 425-447.
- Ashari, N., Koh, H.C., Tan, S.L. and Wong, W.H. (1994), "Factors affecting income smoothing among listed companies in Singapore", *Accounting and Business Research*, Vol. 24 No. 96, pp. 291-301.
- Ayoib, C.A. and Nooriha, M. (2009), "Board independence, ownership structure, audit quality and income smoothing activities: a study on Malaysian market", *Journal of Modern Accounting* and Auditing, Vol. 5 No. 11, pp. 1-13.
- Balsam, S., Krishnan, J. and Yang, J.S. (2003), "Auditor industry specialization and earnings quality", Auditing: A Journal of Practice & Theory, Vol. 22 No. 2, pp. 71-97.
- Barnea, A., Ronen, J. and Sadan, S. (1976), "Classificatory smoothing of income with extraordinary items", *The Accounting Review*, Vol. 56 No. 1, pp. 110-122.
- Behar, A. and Manners, P. (2010), "Distance to growing markets and Sub-Saharan African exports", African Development Review, Vol. 22 No. 2. pp. 316-330.
- Beidleman, C. (1973), "Income smoothing: the role of management", *The Accounting Review*, Vol. 48 No. 4, pp. 653-667.
- Berenson, M.L., Levine, D.M. and Krehbiel, T.C. (2006), *Basic Business Statistics: Concepts and Applications*, Prentice Hall, Upper Saddle River, NJ.
- Cahan, S., Liu, G. and Sun, J. (2008), "Investor protection, income smoothing, and earnings informativeness", *Journal of International Accounting Research*, Vol. 7 No. 1, pp. 1-24.
- Chen, S., Sun, S.Y.J. and Wu, D. (2010), "Client importance, institutional improvements and audit quality in china: an office and individual auditor level analysis", *The Accounting Review*, Vol. 85 No. 1, pp. 127-158.
- Darayseh, M., Gence, I., Abu Al Foul, B. and Abdallah, W., (2010), "The accuracy of the annual financial report data: an empirical analysis of GCC firms", *International Business & Economics Research Journal*, Vol. 9 No. 8, pp. 15-20.
- DeAngelo, L. (1986), "Accounting numbers as market value substitutes: a study of managerial buyouts of public stockholders", *The Accounting Review*, Vol. 61 No. 3, pp. 400-420.
- Dechow, P., Hutton, A. and Sloan, R. (1995), "Detecting earnings management", *The Accounting Review*, Vol. 70 No. 2, pp. 193-225.
- Dechow, P.M. (1994), "Accounting earnings and cash flows as measures of firm performance: the role of accounting accruals", *Journal of Accounting and Economics*, Vol. 18 No. 1, pp. 3-42.



321

Evidence from GCC markets

JAEE 5.3	DeFond, M. and Park, C. (1997), "Smoothing income in anticipation of future earnings", <i>Journal of</i> <i>Accounting and Economics</i> , Vol. 23 No. 2, pp. 115-139.
0,0	Defond, M.L. and Jiambalvo, J. (1994), "Debt covenant violation and manipulation of accruals", <i>Journal of Accounting and Economics</i> , Vol. 17 Nos 1-2, pp. 145-176.
000	Eckel, N. (1981), "The income smoothing hypothesis revisited", <i>Abacus</i> , Vol. 17 No. 1, pp. 28-40.
322	Gassen, J., Fuelbier, R.U. and Sellhorn, T. (2006), "International differences in conditional conservatism: the role of unconditional conservatism and income smoothing", <i>European Accounting Review</i> , Vol. 15 No. 4, pp. 527-564.
	Gordon, M.J. (1966), "Discussion of the effects of alternative accounting rules for nonsubsidiary investments", <i>Journal of Accounting Research</i> , Vol. 4 No. 3. pp. 220-223.
	Hall, C. (2012). "Saudi bourse allows foreign companies to list", <i>Financial Times</i> , January 23, available at: www.FT.COM (accessed July 24, 2014).
	Hall, S. and Frickel, B. (2008), "Income smoothing and executive compensation", <i>University of Nebraska at Kearney</i> , Vol. 3 No. 2, pp. 12-20.
	Hammoudeh, S. and Choi, K. (2006), "Behavior of GCC stock markets and impacts of US oil and financial markets", <i>Research in International Business and Finance</i> , Vol. 20 No. 1, pp. 22-44.
	Healy, P.M. (1985), "The effect of bonus schemes on accounting decisions", <i>Journal of Accounting</i> <i>and Economics</i> , Vol. 7 Nos 1-3, pp. 85-107.
	Hepworth, S.R. (1953), "Periodic income smoothing", <i>The Accounting Review</i> , Vol. 28 No. 1, pp. 32-39.
	Horwtiz, B. (1977), "Comment on income smoothing a review by J. Ronen, S. Sadan, and C. Snow", <i>Accounting Journal</i> , Vol. 1 No. 3, pp. 27-29.
	Hunt, A., Moyer, S. and Shevlin, T. (2000), "Earnings volatility, earnings management, and equity value", working paper, University of Washington, Washington, DC.
	Jones, J. (1991), "Earnings management during import relief investigations", Journal of Accounting Research, Vol. 29 No. 2, pp. 193-228.
	Kaplan, R.S. (1985), "Comments on Paul Healy: the effect of bonus schemes on accounting decisions", <i>Journal of Accounting and Economics</i> , Vol. 7 Nos 1-3, pp. 109-113.
	Keller, G. (2005), Statistics for Management and Economics, Thomson Brooks/Cole, CA.
	Kothari, S. (1992), "Price-earnings regressions in the presence of prices leading earnings: earnings level versus change specifications and alternative deflators", <i>Journal of Accounting and Economics</i> , Vol. 15 Nos 2-3, pp. 173-302.
	Kothari, S.P. (2001), "Capital markets research in accounting", Journal of Accounting and Economics, Vol. 31 Nos 1-3, pp. 105-231.
	Krishnan, J., Sami, H. and Zhang, Y. (2005), "Does the provision of nonaudit services affect investor perceptions of auditor independence?", <i>Auditing: A Journal of Practice & Theory</i> , Vol. 24 No. 2, pp. 111-135.
	Lambert, R. (1984), "Income smoothing as rational equilibrium behavior", <i>The Accounting Review</i> , Vol. 41 No. 4, pp. 604-618.
	Leuz, C., Nanda, D. and Wysocki, P. (2003), "Earnings management and investor protection: an international comparison", <i>Journal of Financial Economics</i> , Vol. 69 No. 3, pp. 505-527.
	Li, S. and Richie, N. (2009), "Income smoothing and the cost of debt", available at: http://ssrn.com/ abstract=1621760 (accessed March 7, 2011).
لاستشارات	المنارخ

www.

- Marashdeh, H.A. and Shrestha, M.B. (2010), "Stock market integration in the GCC countries", Evidence from International Research Journal of Finance and Economics, Vol. 5 No. 37, pp. 102-114.
- Marra, A., Mazzola, P. and Prencipe, A. (2011), "Board monitoring and earnings management pre- and post-IFRS", *International Journal of Accounting*, Vol. 46 No. 2, pp. 205-230.
- Michelson, S.E., Jordan-Wagner, J. and Wootton, C.W. (2000), "The relationship between the smoothing of reported income and risk-adjusted returns", *Journal of Economics and Finance*, Vol. 24 No. 2, pp. 141-159.
- Ohlson, J. and Shroff, P. (1992), "Changes versus levels in earnings as explanatory variables for returns: some theoretical considerations", *Journal of Accounting Research*, Vol. 30 No. 2, pp. 210-226.
- Saidi, N. (2011), "Corporate governance in the GCC: what has been done & what remains", *Qatar Business Review*, November, pp. 11-13.
- Sankar, M. and Subramanyam, K.R. (2001), "Reporting discretion and private information communication through earnings", *Journal of Accounting Research*, Vol. 39 No. 2, pp. 365-386.
- Sbeiti, W. (2008), "Stock markets dynamics, financial sector development and corporate capital structure in the GCC countries", PhD thesis, Durham Business School, University of Durham, Durham.
- Subramanyam, K.R. (1996), "The pricing of discretionary accruals", Journal of Accounting and Economics, Vol. 22 Nos 1-3, pp. 249-281.
- Teoh, S.H. and Wong, T.J. (1993), "Perceived auditor quality and the earnings response coefficient", *The Accounting Review*, Vol. 68 No. 2, pp. 346-366.
- Trueman, B. and Titman, S. (1988), "An explanation for accounting income smoothing", *Journal of Accounting Research*, Vol. 26 No. 3, pp. 127-139.
- Tseng, L.J. and Wen Lai, C. (2007), "The relationship between income smoothing and company profitability: an empirical study", *International Journal of Management*, Vol. 24 No. 4, pp. 727-733.
- Tucker, J. and Zarowin, P. (2006), "Does income smoothing improve earnings informativeness", Accounting Review, Vol. 81 No. 1, pp. 251-270.
- Vuong, Q.H. (1989), "Likelihood ratio tests for model selection and non-nested hypotheses", *Econometrica*, Vol. 57 No. 2, pp. 307-333.
- Wang, D. (2006), "Founding family ownership and earnings quality", Journal of Accounting Research, Vol. 44 No. 3, pp. 619-656.
- White, H. (1980), "A heteroskedasticity consistent covariance matrix estimator and a direct test for heterskedasticity", *Econometrica*, Vol. 48 No. 4, pp. 817-838.
- Wysocki, P. (2004), "Discussion of ultimate ownership, income management, and legal and extra-legal institutions", *Journal of Accounting Research*, Vol. 42 No. 2, pp. 463-474.
- Zawya (2012), "Money, Arab News", March 08, Panel suggests new name for GCC, available at: www.zawya.com

Further reading

- Zughaibi and Kabbani Financial Consultants (2008), *Gulf Investment Guide*, 6th ed., Zughaibi and Kabbani Financial Consultants, Jeddah.
- Zughaibi and Kabbani Financial Consultants (2009), *Gulf Investment Guide*, 7th ed., Zughaibi and Kabbani Financial Consultants, Jeddah.



323

JAEE About the author 5,3 Professor Moade Fa

324

Professor Moade Fawzi Shubita holds a PhD in Accounting and Finance from the University of Manchester (AACSB accredited) in the UK. This bio outlines Moade Shubita's profile up to the acceptance of the paper. Upon completing his studies in 1999, he was appointed Research Associate at the University of Manchester. Later, Moade chaired the Accounting Department at the United Arab Emirates University. He is AQM on the AACSB scale and has held the following posts at the New York Institute of Technology: Assistant Dean in Amman and Bahrain campuses; Associate Campus Dean/Acting Campus Dean in Bahrain Campus; Regional Associate Dean for the School of Management in the Middle East; and Executive Global Assessment Officer for the School of Management. Moade has authored, co-authored, edited and peer-reviewed many publications, and his research interests lie mainly in market-based accounting research. He has gained academic experience in British, American and Middle Eastern systems and has worked in various countries. Moade was promoted to a Full Professor in Accounting on May 17, 2013 by the New York Institute of Technology; confirmed by the Board of Trustees in New York. Professor Moade Fawzi Shubita can be contacted at: moade_shubita@hotmail.com



Reproduced with permission of copyright owner. Further reproduction prohibited without permission.

